Author’s response to reviews

Title: EMOTIONAL IMPACT ON THE RESULTS OF BRCA1 AND BRCA2 GENETIC TEST: AN OBSERVATIONAL RETROSPECTIVE STUDY

Authors:

Sara Mella (smella@cro.it)

Barbara Muzzatti (bmuzzatti@cro.it)

Riccardo Dolcetti (rdolcetti@cro.it)

Maria Antonietta Annunziata (annunziata@cro.it)

Version: 2 Date: 28 Aug 2017

Author’s response to reviews:

Reviewer #1

1. We improved the English language within the entire manuscript.

We confirm that all author details on the revised version are correct, that all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data.